Author’s response to reviews

Title: Injury death certificates without specification of the circumstances leading to the fatal injury - the Norwegian Cause of Death Registry 2005-2014

Authors:

Christian Lycke Ellingsen (christian.lycke.ellingsen@lyse.net)
Marta Ebbing (marta.ebbing@fhi.no)
G. Cecilie Alfsen (g.cecilie.alfsen@medisin.uio.no)
Stein Emil Vollset (stein.emil.vollset@fhi.no)

Version: 2 Date: 24 May 2018

Author’s response to reviews:

Re: Decision letter POHM-D-17-00128R1

We are very pleased that the reviewers are satisfied with the revision of the manuscript.

Regarding the two additional comments:

1. On the use of "intentional/unintentional injuries", we have rephrased two sentences in the introduction, line 135-139.
2. The grammatical error in line 165-166 has been corrected.

The changes are marked with blue colour in the manuscript.

Sincerely,

Christian Lycke Ellingsen

Corresponding author